Author’s response to reviews

Title: Assessment of intracranial pressure with ultrasonographic retrobulbar optic nerve sheath diameter measurement

Authors:

Dachuan Liu (ldc99@sina.com)
Zhen Li (zhenli5188@hotmail.com)
Xuxiang Zhang (zhangxuxiang@vip.163.com)
Liping Zhao (547055672@qq.com)
Jianping Jia (jiajp@vip.126.com)
Fei Sun (sf69@163.com)
Yaxing Wang (narvita@gmail.com)
Daqing Ma (d.ma@imperial.uk)
Wenbin Wei (weiwenbintr@163.com)

Version: 2 Date: 13 Jul 2017

Author’s response to reviews:

Dear Editor,

We are glad to receive your letter. The change that we have made has been highlighted in yellow.

Q: As suggested by reviewer 1 (Q5) please provide the information about the time interval between ultrasonographic measurement and lumbar puncture in the manuscript.

A: We add this information in according part. (see page 4 yellow highlighted)
Best wishes!

Yours sincerely,

Zhen Li and Dachuan Liu

On behalf of all authorship.